Author's response to reviews

Title: Multilevel analysis of caries in underprivileged adolescents of brazilian southeast

Authors:

Fabiana L Vazquez (fabilivazquez@gmail.com)
Karine L Cortellazzi (karinecortellazzi@gmail.com)
Fábio L Mialhe (mialhe@fop.unicamp.br)
Gláucia MB Ambrosano (glaucia@fop.unicamp.br)
Elaine PS Tagliaferro (epstag@gmail.com)
Luciane M Guerra (lumiranda1302@gmail.com)
Armando k Kaieda (armandokaieda@hotmail.com)
Marcelo C Meneghim (meneghim@fop.unicamp.br)
Antonio C Pereira (apereira111@gmail.com)

Version: 2 Date: 23 April 2014

Author's response to reviews: see over
Cover letter

Editorial requests:

1- Title Page: Email addresses of all authors, and the full postal address of the submitting author.
2- Competing Interests: The authors declare that they have no competing interests
3- Financial competing interests

In the past five years I have not received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future. Not is such an organization financing this manuscript (including the article-processing charge).

I don’t hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future.

I don’t hold or currently applying for any patents relating to the content of the manuscript. I don’t have received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript.

I don’t have any other financial competing interests.

4- Non-financial competing interests

There aren’t any non-financial competing interests (political, personal, religious, ideological, academic, intellectual, commercial or any other) to declare in relation to this manuscript.

5- Authors’ Contributions

VFL, CKL, KAK and PAC have made substantial contributions to conception and design, acquisition of data, and analysis and interpretation of data. AGMB participated in the design of the study and performed the statistical analysis. MFL, TESP, GLM and MCM participated in the drafting the manuscript and revising it critically for important intellectual. PAC, TESP and GLM participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.

6- Acknowledgements

The authors wish to acknowledge the financial support from FAPESP (nonprofit; grants #2011/03657-5 and #2011/01768-4). We also give special thanks to the principals of the schools, teachers and all the adolescents who contributed to the accomplishment of the survey.

7- Line Numbering

8- Tables in the main manuscript file after the figure legends and references.